Author's response to reviews

Title: Geriatric rehabilitation of stroke patients in nursing homes: a study protocol

Authors:

Monica Spruit- van Eijk (m.spruit-vaneijk@elg.umcn.nl)
Bianca I Buijck (b.buijck@elg.umcn.nl)
Sytse U Zuidema (s.zuidema@elg.umcn.nl)
Frans LM Voncken (frans.voncken@zorgboog.nl)
Alexander CH Geurts (a.geurts@reval.umcn.nl)
Raymond TCM Koopmans (r.koopmans@elg.umcn.nl)

Version: 2 Date: 2 March 2010

Author's response to reviews: see over
Dear Madam, Sir,

We enclosed our revised study protocol ‘Geriatric Rehabilitation of Stroke Patients in Nursing Homes’ by Monica Spruit- van Eijk, Bianca I. Buijck, Sytse U. Zuidema, Frans L.M. Voncken, Alexander C.H. Geurts, and Raymond T.C.M. Koopmans for consideration of publication in ‘BMC Geriatrics’.

We have revised our manuscript and included the statement regarding the fact that our ethics committee determined that ethics approval was not needed for this study.

In the email received on the 23\textsuperscript{th} of February 2010 there is a remark about following journal style. We have followed the journal template from the website. If this was not correct, could you point out the right one to us.

We are looking forward to your reply.

Kind regards,

Monica Spruit- van Eijk, elderly care physician
Department of primary and community care
117 VPHG Radboud University Nijmegen, Medical Centre
PO Box 9101
6500 HB Nijmegen The Netherlands
e-mail: M.Spruit-vanEijk@elg.umcn.nl
telephone: +31 24 3618834
fax: +31 24 3619558